# Early morbidities following paediatric cardiac surgery: a mixed-methods study

Katherine L Brown,<sup>1</sup>\* Christina Pagel,<sup>2</sup> Deborah Ridout,<sup>3</sup> Jo Wray,<sup>1</sup> Victor T Tsang,<sup>1</sup> David Anderson,<sup>4</sup> Victoria Banks,<sup>1</sup> David J Barron,<sup>5</sup> Jane Cassidy,<sup>5</sup> Linda Chigaru,<sup>1</sup> Peter Davis,<sup>6</sup> Rodney Franklin,<sup>7</sup> Luca Grieco,<sup>2</sup> Aparna Hoskote,<sup>1</sup> Emma Hudson,<sup>8</sup> Alison Jones,<sup>5</sup> Suzan Kakat,<sup>1</sup> Rhian Lakhani,<sup>4</sup> Monica Lakhanpaul,<sup>3,9</sup> Andrew McLean,<sup>10</sup> Steve Morris,<sup>8</sup> Veena Rajagopal,<sup>1</sup> Warren Rodrigues,<sup>10</sup> Karen Sheehan,<sup>6</sup> Serban Stoica,<sup>6</sup> Shane Tibby,<sup>4</sup> Martin Utley<sup>2</sup> and Thomas Witter<sup>4</sup>

- <sup>1</sup>Heart and Lung Division, Great Ormond Street Hospital NHS Foundation Trust, London, UK
- <sup>2</sup>Clinical Operational Research Unit, University College London, London, UK
- <sup>3</sup>Population, Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK
- <sup>4</sup>Departments of Paediatric Intensive Care, Cardiology and Cardiac Surgery, Evelina London Children's Hospital, London, UK
- <sup>5</sup>Departments of Intensive Care and Paediatric Cardiac Surgery, Birmingham Children's Hospital, Birmingham, UK
- <sup>6</sup>Departments of Intensive Care and Paediatric Cardiac Surgery, Bristol Royal Hospital for Children, Bristol, UK
- <sup>7</sup>Paediatric Cardiology Department, Royal Brompton and Harefield NHS Foundation Trust, London, UK
- <sup>8</sup>Department of Applied Health Research, University College London, London, UK
  <sup>9</sup>Community Child Health, UCL Great Ormond Street Institute of Child Health, London, UK
- <sup>10</sup>Department of Intensive care, Royal Hospital for Children, Glasgow, UK

\*Corresponding author Katherine.brown@gosh.nhs.uk

**Declared competing interests of authors:** Katherine L Brown is a member of the Health Technology Assessment (HTA) Clinical Trials Board (2017–21) and a member of the domain expert group of the National Congenital Heart Diseases Audit (2014–19). David L Barron is a member of the National Congenital Heart Disease Audit Steering Committee (2014–18). Monica Lakhanpaul is part of the following boards or panels: HTA Maternal, Neonatal and Child Health (MNCH) Methods Group, HTA MNCH Panel (2012–17) and Psychological and Community Therapies Panel (2012–15). Steve Morris has been a member of the following boards or panels: Health Services and Delivery Research (HSDR) Board Members (2014–18), HSDR Commissioned Board Members, HSDR Evidence Synthesis Sub Board 2016 and the Public Health Research Research Funding Board (2011–17). Thomas Witter was a member of the National Congenital Heart Disease Audit Steering Committee (2014–18).

Published July 2020 DOI: 10.3310/hsdr08300

## **Plain English summary**

Early morbidities following paediatric cardiac surgery Health Services and Delivery Research 2020; Vol. 8: No. 30 DOI: 10.3310/hsdr08300

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

Over 5000 heart surgery procedures are performed in children in the UK each year and early survival has improved to > 98%. Although early survival is important, most people now think that this does not provide enough information. Therefore, we aimed to study health-related issues following surgery that are generally considered as 'bad for the patient'. The medical term for these issues is 'morbidities', but often they are called 'complications'.

We reviewed existing literature, ran three family focus groups and undertook a family online discussion forum. A group of health professionals and family representatives then considered the possible 'morbidities' and decided on a list of the nine most important ones, which were defined by a group of specialists. This included medical complications, such as needing support with a mechanical pump after the operation, but also 'morbidities' that were particularly serious to families, such as feeding difficulties.

Working in five of the UK centres, we involved children aged < 17 years with congenital heart disease and their parents to:

- successfully test a new method for nurses to check child development in 1200 children (we found that this worked in children aged between 4 months and 5 years old)
- measure how often 'morbidities' happened after 3090 operations (we found that 21.8% experienced at least one)
- measure the impact of 'morbidities' in 666 children over 6 months (one important finding was that although patient quality of life was reduced when these happened, over 6 months it improved in survivors).

We developed new ways to inform parents and the organisation that monitors the results of children's heart surgery (national audit) about complications, with a view to empowering parents. Monitoring 'morbidity' is more difficult than monitoring mortality, so it requires resources and clinician commitment. Further research is needed to work out how best to help children with feeding problems and delays in development.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Brown et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

## **Health Services and Delivery Research**

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### HS&DR programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-servicesand-delivery-research.htm

#### This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 12/5005/06. The contractual start date was in January 2014. The final report began editorial review in August 2018 and was accepted for publication in May 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Brown *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

## **NIHR Journals Library Editor-in-Chief**

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk